Author's response to reviews

Title: Taste hyposensitivity in Japanese schoolchildren.

Authors:

Mari Ohnuki (ohnuki.ohp@tmd.ac.jp)
Ueno Masayuki (ueno.ohp@tmd.ac.jp)
Takashi Zaitsu (zaitsu.ohp@tmd.ac.jp)
Yoko Kawaguchi (yoko.ohp@tmd.ac.jp)

Version: 3 Date: 31 March 2014

Author's response to reviews: see over
March 31, 2014

Dr Wael Sabbah
Editor in Chief,
BMC Oral Health
MS 1580437166115066

Mari Ohnuki, Masayuki Ueno, Takashi Zaitsu & Yoko Kawaguchi
"Taste hyposensitivity in Japanese schoolchildren"

Dear Dr Wael Sabbah
Editor in Chief,
BMC Oral Health,

Thank you very much for your e-mail dated March 15, 2014. We are very pleased that you expressed an interest in our paper.

Response to Reviewer
We have edited sentences in abstract, methods and discussion. And we have rephrased last sentence in conclusion.

We hope that the revised manuscript is now suitable for publication in BMC Oral Health.

Sincerely yours,
Mari Ohnuki DDS, PhD